Author’s response to reviews

Title: INCREASED INCIDENCE OF SUSAC SYNDROME: A CASE SERIES STUDY

Authors:

Adi Wilf-Yarkoni (adi.wilf@gmail.com)
O. Elkayam (orie@tlvmc.gov.il)
O. Aizenstein (lehavio@netvision.net.il)
O. Yahav (yahavo@tlvmc.gov.il)
V. Furer (vikushf@gmail.com)
D. Zur (dinahzur@gmail.com)
M. Goldstein (michaellag@tlvmc.gov.il)
D. Barequet (danabarequet@gmail.com)
H. Hallevi (Hen.hallevi@gmail.com)
A. Karni (aronkarni@yahoo.com)
Z Habot-Wilner (zwilner@hotmail.com)
K Regev (regevke@gmail.com)

Version: 3 Date: 14 Aug 2020

Author’s response to reviews:

We have made the requested formatting changes to the manuscript.

Dr. Adi Wilf-Yarkoni